HO 3

CSC 354S

SYSTEMS MODELLING AND DISCRETE SIMULATION

SPRING 1997

UNIVERSITY OF TORONTO

PROBLEM SET 2

Purpose : To get you to week 9 in the course without having designed and run a full-scale simulation program.

General Comments:

Specific Comments : The electronic submission for this problem set must be done by 2:00 p.m.

on Thursday, March 13. The hard-copy handin for this problem set is due by in tutorial on

Thursday, March 13. Hand in hard-copy answers in some secure package; specific handin requirements may be posted on the newsgroup. Remember to hand in a signed equality sheet with your hard-copy.

1. Question 10.14 from BCN

2. Question 10-18 from BCN

3. Question 10-21 from BCN

4. Question 10-22 from BCN

5. Question 10-23 from BCN

6. Question 10-26 from BCN